
Status 19 Nov
Today's status:

In general, pretty good by now so John and Ling return today.  (Kurt remains alone for a bit longer.)  Kryptons are being recalibrated in Boulder with shorter 
pathlengths and will be tested through serializers retrieved (today) from Manitou.  They will be installed by Tom and Gordon just prior to operations 29/30 
Nov.  Fixes to the SPN1 and TP01 should also be straightforward.  Problems with Tsoils seem to be caused by inadequate conformal coating of the 
boards.  We'll probably just apply nail polish to fix.

Site Battery 
&Solar

Sonic Krypton Baro TRH Rad Soil Soil.aux Precip Shadowband Prop

1 ok ok not installed ok ok ok ok some Tsoil bad not 
planned

SPN1 in Boulder
(A/D found bad)
Licor ok

ok

2 lowest ok not installed ok ok ok ok not planned not 
planned

not planned not 
planned

3 ok ok  not installed ok ok ok ok not planned ok not planned not 
planned

4  ok ok not installed ok ok ok ok not planned ok not planned not 
planned

5 ok ok  not installed ok ok ok ok Tsoil ok
 Qsoil erratic

Gsoil ok  
TP01in Boulder
(will be reprogrammed)

not 
planned

not planned not 
planned

6 ok ok  not installed ok ok ok Tsoil profile kinked ok ok not planned ok

7 ok ok  not installed ok ok ok ok not planned not 
planned

ok not 
planned
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